Author's response to reviews

Title: Establishing a Nationwide Emergency Department-based Syndromic Surveillance System for Better Public Health Responses in Taiwan

Authors:

Tsung-Shu Joseph Wu (wcsg@msn.com)
Fuh-Yuan Frank Shih (fystone@mail2000.com.tw)
Muh-Yong Yen (myyen1121@gmail.com)
Jiunn-Shyan Julian Wu (julianwu77@hotmail.com)
Shiou-Wen Lu (lsw@cdc.gov.tw)
Chao Hsiung (hsiung@nhri.org.tw)
Kevin Chi-Ming Chang (kevin@cdc.gov.tw)
Jr-How Chou (jchou@cdc.gov.tw)
Yu-Tseng Chu (littleppiiiggg@yahoo.com.tw)
Hang Chang (emergency@ms.skh.org.tw)
Chan-Hsien Chiu (zschou@cdc.gov.tw)
Fu-Chiang Richard Tsui (tsui@cbmi.pitt.edu)
Michael M. Wagner (mmw@cbmi.pitt.edu)
Ih-Jen Su (suihjen@nhri.org.tw)
Chwan-Chuen King (cc_king99@hotmail.com)

Version: 9 Date: 11 December 2007

Author's response to reviews:

Dear Editors,

We are pleased to resubmit this revised manuscript entitled “Establishing a Nationwide Emergency Department-based Syndromic Surveillance System for Better Public Health Responses in Taiwan.”

As we know, the past three decades have documented increases worldwide in severe outbreaks of emerging/re-emerging infectious diseases with unknown etiological agents. Moreover, global travel and commerce provide ever-increasing opportunities for infectious diseases to cross national borders. Much has been published on the use of information technology to establish syndromic surveillance systems to detect emerging outbreaks and other public health threats. Our paper, however, describes the first such system to be established in a non-English-speaking Asian country. This paper traces the process, from planning to implementation, of building a nationwide syndromic surveillance system in Taiwan and addresses major practical concerns and barriers encountered. We would be very glad to share our valuable experience with readers interested in this public health issue of global concern.

Thus far, the Asian countries most affected by frequent outbreaks of H5N1 have included Mainland China, Vietnam, Thailand, and Indonesia. The high volume of travel between these countries and Taiwan raises the concern that a
human-to-human pandemic, should it occur, may spread rapidly to Taiwan and cause considerable loss of life. As surveillance is the most effective prevention and control measure against avian influenza recommended by the World Health Organization, our goal has been to institutionalize a means of monitoring various syndrome groups and disease patterns (including respiratory, gastrointestinal, etc.) for trends and indicators. This information makes it possible for us to detect potential outbreaks early and provide valuable opportunities for preparation and preemption.

We've revised this manuscript with English reviewer and changed the required modification from reviewers opinions. We hope this version would be much better than previous.

Your consideration and kind assistance are greatly appreciated.

Sincerely Yours,

Mr. Tsung-Shu Joseph Wu and Dr. Chwan-Chuen King